

Trends, Breaks, and Biases in the Frequency of Reported Glacier Lake Outburst Floods

Georg Veh, Natalie Lützow, Varvara Kharlamova, Dmitry Petrakov, Romain

Hugonnet, Oliver Korup

▶ To cite this version:

Georg Veh, Natalie Lützow, Varvara Kharlamova, Dmitry Petrakov, Romain Hugonnet, et al.. Trends, Breaks, and Biases in the Frequency of Reported Glacier Lake Outburst Floods. Earth's Future, 2022, 10, 10.1029/2021EF002426 . insu-03671379

HAL Id: insu-03671379 https://insu.hal.science/insu-03671379

Submitted on 18 May 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License

f





RESEARCH ARTICLE

10.1029/2021EF002426

Key Points:

- We present a global inventory of almost 2,000 glacier lake outburst floods (GLOFs) for the period 1901–2017
- Only a few regions had a growing number of reported GLOFs, likely more linked to increasing research activity than to atmospheric warming
- We estimate that on average two to four out of five GLOFs might have escaped human notice in the early to mid-20th century

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

G. Veh, georg.veh@uni-potsdam.de

Citation:

Veh, G., Lützow, N., Kharlamova, V., Petrakov, D., Hugonnet, R., & Korup, O. (2022). Trends, breaks, and biases in the frequency of reported glacier lake outburst floods. *Earth's Future*, *10*, e2021EF002426. https://doi. org/10.1029/2021EF002426

Received 13 SEP 2021 Accepted 28 FEB 2022

Author Contributions:

Conceptualization: Georg Veh, Oliver Korup

Data curation: Georg Veh, Natalie Lützow, Varvara Kharlamova, Dmitry Petrakov, Romain Hugonnet Formal analysis: Georg Veh Investigation: Georg Veh Methodology: Georg Veh, Oliver Korup Resources: Oliver Korup Software: Georg Veh Supervision: Oliver Korup

© 2022 The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Trends, Breaks, and Biases in the Frequency of Reported Glacier Lake Outburst Floods

Georg Veh¹, Natalie Lützow¹, Varvara Kharlamova², Dmitry Petrakov², Romain Hugonnet^{3,4,5}, and Oliver Korup^{1,6}

¹Institute of Environmental Science and Geography, University of Potsdam, Potsdam, Germany, ²Faculty of Geography, Moscow State University, Moscow, Russia, ³Université de Toulouse, CNRS, LEGOS, Toulouse, France, ⁴Laboratory of Hydraulics, Hydrology and Glaciology (VAW), ETH Zürich, Zürich, Switzerland, ⁵Swiss Federal Institute for Forest, Snow and Landscape Research (WSL), Birmensdorf, Switzerland, ⁶Institute of Geosciences, University of Potsdam, Potsdam, Germany

Abstract Thousands of glacier lakes have been forming behind natural dams in high mountains following glacier retreat since the early 20th century. Some of these lakes abruptly released pulses of water and sediment with disastrous downstream consequences. Yet it remains unclear whether the reported rise of these glacier lake outburst floods (GLOFs) has been fueled by a warming atmosphere and enhanced meltwater production, or simply a growing research effort. Here we estimate trends and biases in GLOF reporting based on the largest global catalog of 1,997 dated glacier-related floods in six major mountain ranges from 1901 to 2017. We find that the positive trend in the number of reported GLOFs has decayed distinctly after a break in the 1970s, coinciding with independently detected trend changes in annual air temperatures and in the annual number of field-based glacier surveys (a proxy of scientific reporting). We observe that GLOF reports and glacier surveys decelerated, while temperature rise accelerated in the past five decades. Enhanced warming alone can thus hardly explain the annual number of reported GLOFs, suggesting that temperature-driven glacier lake formation, growth, and failure are weakly coupled, or that outbursts have been overlooked. Indeed, our analysis emphasizes a distinct geographic and temporal bias in GLOF reporting, and we project that between two to four out of five GLOFs on average might have gone unnoticed in the early to mid-20th century. We recommend that such biases should be considered, or better corrected for, when attributing the frequency of reported GLOFs to atmospheric warming.

Plain Language Summary Glacier lakes have been growing rapidly following atmospheric warming, glacier retreat, and the exposure of new storage space for meltwater. Many of these lakes have unstable dams and are exposed to impacts from rock/ice avalanches, raising concerns of a commensurate increase of glacier lake outburst floods (GLOFs). Their repeated catastrophic impacts motivate robust assessments that examine the extent to which the frequency of reported GLOFs has changed under ongoing atmospheric warming. Collating 2,000 cases between 1901 and 2017, we find little evidence for changes in the annual number of reported GLOFs, however, though air temperatures have increased markedly since the 1970s. Accordingly, the temperature-driven increase in glacier lake area and number has unclear links with the frequency of reported GLOFs. Only glacier-dammed lakes have had more reported outburst floods over the past five decades, as the stability of dams may have declined due to continued glacier downwasting. Changes in research activity have likely biased the actual regional rate of GLOFs, and we estimate that many hundreds of GLOFs went unnoticed in the early 20th century. We invite future research to learn more about the physical drivers of regional GLOF trends to improve projections of GLOF occurrence under ongoing atmospheric warming.

1. Introduction

Ongoing atmospheric warming has been a key driver of glacier retreat in high mountains in past decades (Marzeion et al., 2014). Shrinking glacier volumes contribute to sea-level rise, change the freshwater availability in mountain rivers and pose hazards to mountain livelihoods (Huss et al., 2017; Milner et al., 2017; Zemp et al., 2019). Among the most publicized ice-related hazards are glacier lake outburst floods (GLOFs) (Emmer, 2018). These floods emerge from glacier lakes that store meltwater behind a glacier or moraine dam, or in over-deepened parts of exposed glacier beds. Other lakes form on top of glaciers in supraglacial ponds, below glaciers in subglacial lakes,



Validation: Georg Veh Visualization: Georg Veh Writing – original draft: Georg Veh, Oliver Korup Writing – review & editing: Georg Veh, Dmitry Petrakov, Romain Hugonnet, Oliver Korup or within the ice body as englacial water pockets (Iturrizaga, 2011). Among these dam types, glacier-dammed lakes have dominated in previous inventories of GLOFs (Carrivick & Tweed, 2016). Ice-dammed lakes can fill and burst out repeatedly, compared to moraine dams that mostly burst once as their dams are large destroyed during failure. Some hundreds of GLOF have occurred in past centuries, meeting communities along mountain channels largely unprepared (Carrivick & Tweed, 2016). GLOFs claimed >12,500 fatalities, loss of livestock, farmland, and infrastructure, and disruption of traffic and communication routes since the 1950s (Carrivick & Tweed, 2016). Growing population pressure and economic interests in hydropower generation, forestry, and mining in high mountains call for adapting to and mitigating the impacts from GLOFs (Immerzeel et al., 2020; Schwanghart et al., 2016).

Many scientists argue that the annual number of GLOFs might increase as a consequence of atmospheric warming (Bajracharya & Mool, 2009; Bolch et al., 2012; Clague & Evans, 2000; Cook et al., 2018; Harrison et al., 2018; Liu et al., 2014; Miles et al., 2018; Richardson & Reynolds, 2000; Shugar et al., 2020; Tweed & Russell, 1999; Y. Chen, 2010). This hypothesis is based on the climate-driven retreat of glaciers that began in the European Alps in the 1850s, and few decades later in other mountain regions (Paul & Bolch, 2019). Evidence has grown that high mountains had higher rates of temperature increase than lower elevations (Mountain Research Initiative EDW Working Group, 2015), contributing to an accelerated glacier mass loss in past decades (Hugonnet et al., 2021; Zemp et al., 2019). Thousands of lakes have been forming at the margins of retreating glaciers (Shugar et al., 2020), with reported outbursts deemed to be on the rise, especially since the beginning of the 20th century (Harrison et al., 2018). Rising air temperatures could change the meltwater input to lakes, and thus may increase glacier lake volume, given a suitable geometry of the newly exposed basin (Richardson & Reynolds, 2000). In the Himalayas, for example, only 10% of all glaciers terminate in proglacial lakes; yet some of those lakes may accelerate ice melt and generate more meltwater (King et al., 2019). Warmer air and lake temperatures may thin ice dams and thus lower the threshold to open subglacial drainage channels, likely initiating more GLOFs from glacier-dammed lakes (Clarke, 2003; Kingslake & Ng, 2013; Ng & Liu, 2009). Atmospheric warming might also increase the frequency of moraine-dam failures. Buried ice within a moraine dam can melt, thus reducing cohesion and promoting dam collapse. Gradual settlement of the dam can reduce the lake freeboard and increase the chance for overtopping (Richardson & Reynolds, 2000). Thawing ice and permafrost could also destabilize adjacent glaciers and rockfalls (Haeberli et al., 2017) such that avalanches of ice and rock enter glacier lakes, leading to overtopping and dam breaks (Emmer & Vilímek, 2013; Harrison et al., 2018; Richardson & Reynolds, 2000). A growing number of new lakes due to ongoing glacier retreat will therefore require effective management strategies of this natural hazard, on time scales that may extend well beyond this decade (Haeberli et al., 2017).

Still, attributing the annual rate of GLOFs to atmospheric warming is in its infancy (Harrison et al., 2018). Understanding of physical processes and parameterization of models is often insufficient to reproduce the process chain of climate-driven glacier retreat, lake formation and growth, occurrence of associated triggers, and outburst eventually (Harrison et al., 2018). Local predictions of the occurrence and timing of GLOFs from temperature data alone have been only partly successful (Kingslake & Ng, 2013). Identifying regional links between GLOF frequency and atmospheric warming requires consistent databases in turn. Yet published GLOF inventories cover different regions and time intervals, and GLOF frequencies have been reported as either inconclusive or decreasing (Carrivick & Tweed, 2016; Harrison et al., 2018; Nie et al., 2018), unchanged (Veh et al., 2019), fluctuating (Emmer et al., 2020), or increasing (Iribarren Anacona et al., 2015) in past decades. One caveat for learning GLOF trends is a temporal bias such that entries may be missing systematically in databases due to historic changes in data recording and instrumentation (Veh et al., 2019). The long-term global GLOF record might also be biased geographically towards cases with societal impacts in regions that have long traditions in mountaineering, glacier research, habitation, and infrastructure development (Carrivick & Tweed, 2016; Harrison et al., 2018). These potential biases make it unclear whether the global increase in reported GLOFs (Carrivick & Tweed, 2016) is linked to a warming atmosphere or to a growing research attention with access to increasingly detailed data coverage. Distinct reported breaks in the yearly number of GLOFs in the 1970s (Harrison et al., 2018) and 1990s (Carrivick & Tweed, 2016) likely reflect underreporting, especially as dozens of previously unknown cases were detected only recently (Barka et al., 2020; Emmer, 2017; Jacquet et al., 2017; Veh et al., 2019; Wilson et al., 2018).



Here we test the notion of such a break in the number of yearly reported GLOFs with a major update to the existing catalog of GLOFs. Our goal is to examine whether similar breaks appear in time series of annual air temperatures and proxies of research activity, and how well these predict the trends in annual GLOF counts. In distinguishing between moraine, glacier, and other types of dams, we will estimate the relative contributions of these two predictors to the annual number of reported GLOFs.

2. Materials and Methods

2.1. Data Collection

Carrivick and Tweed (2016) collated the only global GLOF inventory that incorporates floods from all types of glacier lakes. We expand this catalog using continuously updated inventories from public authorities in Alaska, Iceland, and Norway, in addition to other recent regional inventories (Emmer, 2017; Bačka et al., 2020; Bhambri et al., 2019; Haeberli, 1983; McKillop & Clague, 2007; Nie et al., 2018; Veh et al., 2019; Wilson et al., 2018). Besides these databases, our search for GLOFs encompassed a total 721 different sources including analyses of stream gages; satellite and aerial images; stratigraphy; tree rings; reports by local authorities; news outlets; workshop proceedings; social media accounts; and unpublished work. To be included, a GLOF must have a reported date (at least the year of occurrence) or a range of plausible dates; coordinates of the source lake; and at least one reference to enter our database. We distinguish between outbursts from lakes dammed by moraines or by glaciers, given their prevalent role in previous regional records (Carrivick & Tweed, 2016); rarer cases such as supraglacial, bedrock-dammed, or englacial water bodies are grouped as "other" types of dams. Some cases were discarded because they belonged most likely to other types of cryospheric mass flows, mostly in High Mountain Asia where accounts from social media confused GLOFs with debris flows or flash floods after convective storms. We also collated floods from lakes beneath ice-covered volcanoes though we ignore these cases in our analysis as meltwater accumulates by a geothermal interaction of magma with ice and snow, and less so by a climatic control (Gudmundsson et al., 1995; Magnússon et al., 2012). For each dated case, we further documented the mountain range; the country from which the flood has originated; the name of the parent glacier local name and the ID in the Randolph Glacier Inventory (Pfeffer et al., 2014, RGI V6.0); and the name of the burst glacier lake, if available. We also extracted, if possible, summary statistics of lake volume; flood volume; and peak discharge; and also how these diagnostics had been estimated. Finally, we report any impacts; damages; losses; or specific remarks associated with these events.

We use two proxies to assess the competing roles of atmospheric warming and research interest on GLOF reporting. We extracted monthly time series of air temperatures for every GLOF source location from the CRU TS V4.05 data set (Harris et al., 2020). CRU TS spans the period 1901–2020, has a resolution of $0.5^{\circ} \times 0.5^{\circ}$, and was interpolated from weather stations. We aggregated the data from monthly to mean annual temperatures.

To measure annual research interest in GLOFs, we refrained from a bibliometric analysis, as research on GLOFs before 2000 was low: Emmer (2018) found only 170 GLOF research articles listed on the Web of Science between 1979 and 2000. The low publication activity on GLOFs contrasts with the long research tradition on glaciers. Glacier monitoring has been coordinated internationally since 1894 to develop a strategy for quantifying changes in glacier length, area, volume, and mass in response to atmospheric warming (WGMS, 2020). The compilation, analysis, and dissemination of glaciological surveys in annual bulletins followed standardized rules long before regional GLOF inventories became available. Apart from glacier surveys, these bulletins also report on "special events" including many dozens of GLOFs. Glacier surveys can thus serve as a first-order proxy of research activity in glaciated high mountains. We therefore obtained the annual number of *in-situ* glacier front and mass balance measurements in the period 1901–2017 as reported in the Glacier Change Bulletin No. 3 by the World Glacier Monitoring Service (WGMS, 2020).

2.2. Change-Point Models for Global Changes in GLOF Reporting, Temperatures, and Research Activity

Our strategy was to test, first, whether we can objectively locate the previously reported change in GLOF counts in our updated inventory (Carrivick & Tweed, 2016; Harrison et al., 2018) and, second, whether we can locate this break independently also in the time series of air temperatures or research activity. For this purpose, we chose Bayesian piecewise regression models to estimate parameters and their uncertainties, while including previous knowledge. The three piecewise models M_1 , M_2 , and M_3 encode our initial hypotheses of two distinct periods



in the rate of reported GLOFs (M_1), temperature (M_2), or research activity (M_3) in the 20th century. We predict the outcome y (the GLOF rate, temperature or research activity) from two linear trends during the 118-year study period $x = \{1901, 1902, \dots, 2017\}$ that are connected by one change point in year x_c :

$$y_i = \beta_0 + \beta_1 x_i + \beta_2 (x_i - x_c) I(x_i > x_c)$$
(1)

We prefer this model structure to more complex models with multiple change points or non-linear trends to keep the number of estimated model parameters small and the coefficients interpretable. Our model jointly estimates an intercept β_0 and the trends before (β_1) and after (β_2) the change-point x_c . The sign of the parameters β_1 and β_2 indicate positive, negative, or ambiguous (if their posterior interval estimates contain zero with a specified probability) effects on the outcome y. The indicator function $I(\cdot)$ returns 0 or 1, depending on whether its argument is false or not.

To obtain posterior distributions of the model parameters given the data, we defined a likelihood function and specified our prior knowledge for the parameters. Both the annual number of GLOFs and the annual number of glaciological surveys are count data, so that we chose a Poisson likelihood for models M_1 and M_3 . In the Poisson distribution, the rate λ (i.e., the average number of counts per unit time) and the variance are identical, so that we need to estimate only the rate λ_i for the two segments. We formulate the Poisson regression model as

$$n_i \sim \text{Poisson}\left(\lambda_i\right)$$
 (2)

$$\log(\lambda_{i}) = \beta_{0} + \beta_{1}x_{i} + \beta_{2}(x_{i} - x_{c})I(x_{i} > x_{c})$$
(3)

The log-link function ensures that the rate λ_i in either segment is always positive, hence estimating an exponential (rather than a linear) relationship of *n* counts with *x* years. For model M_2 , we assume a normal (Gaussian) likelihood to estimate the mean annual air temperature t_i

$$t_i \sim N(\mu_i, \sigma^2)$$
 (4)

$$\mu_i = \beta_0 + \beta_1 x_i + \beta_2 (x_i - x_c) I(x_i > x_c)$$
(5)

The model assumes a constant noise σ^2 in either segment of the change-point model. Following Bayes' Rule, we multiply the likelihood functions (Equations 2 and 4) with the distributions of all (hyper-)priors, and renormalize this product to obtain a marginal posterior distribution for each parameter. We put robust, though weakly informed, *t*-distributed priors on β_0 , β_1 , β_2 , x_c , and σ (Table S1 in Supporting Information S1). These priors allows for both positive or negative trends given the inconclusive findings in previous regional studies. We truncated the prior distribution on x_c at the lower 5% and upper 95% of observed years *x* to avoid few data points at the boundaries of the data range generating spurious fits. The priors refer to standardized data with mean zero and unit standard deviation to improve sampling efficiency. We numerically approximated the posterior distributions of the parameters (Figure S1 in Supporting Information S1) using a Gibbs sampling algorithm that is implemented in the JAGS software, and called from the package mcp (Lindeløv, 2020) in the statistical software R. We run four parallel chains of 20,000 samples after 4,000 warm-up runs each, and checked for numerical divergences.

We compare the three change-point models against simpler models without a change point, where the term following β_2 in Equation 1 disappears. To this end, we maintained the choice of prior distributions and likelihood functions. We assessed model performance using leave-one-out (LOO) cross-validation implemented in the R package loo (Vehtari et al., 2017). LOO approximates the predictive accuracy of a given model by calculating the log-likelihood from posterior simulations of the parameter values. The resulting expected log predictive density (ELPD) summarizes the predictive error of a Bayesian model (Vehtari et al., 2017).

2.3. Regional Trends in GLOF Reporting in the Period 1973–2017

We interpret the period following the mean change-point location in 1973 as a period of more consistent GLOF reporting, judging from the unabated research effort since then (Figure 1). To learn more about recent trends in GLOF frequency, we used a hierarchical model structure that accounts for differences in reported GLOFs both by dam type d and region r. We chose a nested framework, in which the three different dam types (ice, moraine, and other) encompass the six study regions, resulting in a total of 18 different levels. Splitting the data into smaller groups introduces a higher number of years without GLOF occurrences than we would expect from a Poisson





Figure 1. Comparing the annual number of reported GLOFs with annual air temperatures and research activity in the period 1901–2017. (a) Map of the six study regions. Inset shows the number of GLOFs per decade. (b) Annual number of reported GLOFs. (c) Mean annual air temperature averaged from all sites that produced at least one GLOF. (d) Total number of annual glacier surveys in our study regions. In panels (b–d), thick lines are the means and shades are the 50%–95% posterior predictive highest density intervals (HDIs) from a Bayesian piecewise Poisson regression of the outcome on the *y*-axis versus year. Bottom black lines and probability densities are the prior and posterior location of the changepoint.

distribution. We assumed that any zero counts in the data can arise either from the case that no GLOF had occurred, or that a GLOF occurred without notice or reporting. We chose a zero-inflated Poisson (ZIP) distribution to model the probability $p_{d,r}$ for observing a zero outcome for a given dam type and region. The ZIP is a mixture of a Poisson distribution for cases if the annual number of GLOFs $n_i > 0$ and applies a logit function to model the probability of no GLOF ($n_i = 0$). To learn the conditional GLOF rate λ for a given dam type and region in $x = \{1973, 1974, \dots, 2017\}$ years, we formulate the ZIP model as

$$n_{i|d,r} \sim \text{ZIP}(\lambda_{i|d,r}, p_{d,r}) \tag{6}$$

$$\log(\lambda_{i|d,r}) = \beta_{0,i|d,r} + \beta_{1,i|d,r} x_{i|d,r}$$
(7)

$$logit(p_{d,r}) = \alpha_{d,r}.$$
(8)

This model allows the group-level intercepts β_0 and slopes β_1 to change with time *x*. We assumed that the probability of missing a GLOF remains unchanged between 1973 and 2017, guided by studies that found years without GLOFs in any of the past three decades (Emmer et al., 2022; Veh et al., 2018). Again, we standardized the predictors, and define weakly informed (hyper-)priors for the intercepts, slopes, and the standard deviations that model the variance between the levels (Table S3 in Supporting Information S1). We encoded the multi-level model using the package brms (Bürkner, 2017) in R. Brms builds on the statistical modeling platform Stan that approximates posterior distributions using a No U-Turn (NUTS) algorithm. We run four parallel chains of 2,000 samples after 1,500 warm-up runs, and found no numerical divergences.

2.4. Impacts of Temperatures and Research Activity on GLOF Reporting in the Period 1973–2017

We used the same period (1973–2017) and the same hierarchical model structure to learn more about the impacts of atmospheric warming and research activity on GLOF reporting. We modify the ZIP model from Equations 6–8 to estimate the mean annual GLOF count $\lambda_{i|d,r}$ for a given dam type and region from data pairs of mean annual air temperatures $t_{i|d,r}$ and the \log_{10} -transformed annual number of glaciological surveys $s_{i|d,r}$

$$n_{i|d,r} \sim \text{ZIP}(\lambda_{i|d,r}, p_{d,r}) \tag{9}$$

$$\log\left(\lambda_{i|d,r}\right) = \beta_{0,i|d,r} + \beta_{1,i|d,r} t_{i|d,r} + \beta_{2,i|d,r} s_{i|d,r}$$
(10)

$$logit (p_{d,r}) = \alpha_{d,r} \tag{11}$$

We maintain the choice of weakly informative priors and the setup of sampling design, and report the prior and posterior distributions of the model parameters in Table S4 in Supporting Information S1.

2.5. Hind- and Forecasting GLOF Counts

For either model, we integrated out the parameters to simulate draws from the predictive posterior distribution for unobserved inputs. We can thus hindcast GLOF counts *n* for any given year *x*, or equivalently for any pair of air temperature *t* and number of glaciological surveys *s* in a given year. To this end, we generated 8,000 posterior samples for each level and year. We

stacked these samples such that we obtain a posterior predictive distribution of the total hindcast GLOF count in the period 1901–1972.

To forecast GLOF counts for each group level, we repeated the procedure of learning, and predicting from, hierarchical Poisson regression models. We therefore changed the observation period to $x = \{1901, 1902, ..., 1972\}$, including the mean annual air temperatures *t* and number of glaciological surveys sreported during that period. We used prior distributions identical to those described in Table S3 in Supporting Information S1 (for the model in Equation 6–8 using the predictor "Year") and Table S4 in Supporting Information S1 (for the model in Equation 9–11 using the predictors "Annual air temperature" and "Glaciological surveys"). We maintained the sampling design in brms, that is, four parallel chains of 2,000 samples after 1,500 warm-up runs, to approximate posterior distributions of the model parameters (Figures S4 and S5 in Supporting Information S1). Finally, we generated posterior predictive samples for unobserved inputs for each year during the period 1973–2017, which we stacked to obtain distributions of forecast GLOF counts.

3. Results and Discussion

3.1. Breaks in GLOF Reporting, Air Temperatures, and Research Activity in the 1970s

In reviewing 721 different sources of information, we collated the largest available inventory of dated GLOFs for the Andes, the Pacific Northwest, Iceland, the European Alps, Scandinavia, and High Mountain Asia. We found a total of 1,997 GLOFs that occurred in these six regions between 1901 and 2017 (Figure 1a), thus doubling the count of an earlier global collection (Carrivick & Tweed, 2016). The Pacific Northwest (n = 667; 33%) and High Mountain Asia (n = 479; 24%) had the most reported cases (Figure 1a). We find that the annual reporting rate of GLOFs increased more than fourfold, from $6^{+5}/_{-4}$ GLOFs (posterior median and 95% highest density interval, HDI) in 1901 to $28^{+12}/_{-10}$ GLOFs in 2017. In searching for a clear break in GLOF reporting, we found that a two-piece regression model outperforms a simple regression (Table S2 in Supporting Information S1). The piecewise model has a credible break in the period $1973^{+12}/_{-10}$ (posterior median and 95% HDI), separating a steeper trend in the early 20th century from a flatter, recent one (Figure 1b, Figure S1 in Supporting Information S1). Accordingly, the average trend of +0.21 GLOFs yr⁻¹ from 1901 to 1973 dropped by a quarter to +0.15 GLOFs yr⁻¹ between 1973 and 2017.

To explain these changing trends in GLOF reporting, we chose mean annual air temperature (Harris et al., 2020) and the annual number of surveyed glaciers (WGMS, 2020) as proxies of a climate control of GLOF occurrences versus one influenced mainly by research-driven documentation. Using the same piecewise structure in independent models, we obtain overlapping change-point intervals in the period $1980^{+7/}_{-8}$ for annual air temperatures (Figure 1c), and $1972^{+1}/_{-2}$ for the annual number of glacier surveys (Figure 1d). These time series have opposing trends embracing the change point interval, with contrasting implications on reported GLOF occurrences. We observe a credible warming trend ($+0.003 \pm 0.0025^{\circ}$ C yr⁻¹) before the change-point that accelerated more than ninefold to $+0.028 \pm 0.008^{\circ}$ C yr⁻¹ after. The trends in temperature are thus opposite to the trends in GLOF reporting either side of the break point, so that we observe a decreasing amount of GLOFs per unit temperature increase since the 1970s. In contrast, both GLOF reporting and glaciological surveys (Figure S1 in Supporting Information S1). Such similarities have also occurred in the seven decades before, when both time series have credible positive trends. Thus, the similar distribution of trends and breaks may indicate under-reporting of GLOFs in the early 20th century, assuming links between research activity and GLOF reporting.

3.2. Minor Changes in Regional GLOF Reporting Rates Since the 1970s

All posterior change-point intervals overlap in the early 1970s. In distinguishing further between glacier (72%), moraine (15%), and other dam types (13%, including supraglacial lakes, englacial water pockets, and glacierbed overdeepenings), the hierarchical regression model reveals only few credible trends in GLOF reporting between 1973 and 2017 (Figure 2). Among these are ice-dammed lakes in the Andes, the Pacific Northwest, and Scandinavia, where rapid retreat of tributary glaciers has created space for lakes impounded by trunk glaciers; many of these have burst out repeatedly (Geertsema & Clague, 2005; Jacquet et al., 2017; Wikstrom Jones &





Figure 2. Trends in GLOF reporting in the period 1973–2017. (a) Changes in GLOF reporting rates over time in each region. Thick lines are mean rates and shade is the 95% HDI of reported GLOFs by three different dam types; gray lines and tones are dam types that have no credible trends. (b) Posterior trends (i.e., regression slopes) in GLOF reporting. We standardized the predictor "Year" to measure its relative contribution to the outcome. Asterisks denote posterior distributions that are credibly different from zero (thick vertical line).

Wolken, 2019). High Mountain Asia had no credible trend in ice-dam failures, though at least seven glaciers have surged and dammed temporary lakes that produced multiple GLOFs (Bhambri et al., 2019). In Iceland, we observe a credible decline in reported GLOFs, as nearly all ice-dammed lakes responsible for outbursts in the 20th century are gone today (Guðmundsson et al., 2020). Trends in reported moraine-dam failures remained unchanged in all regions.

3.3. Resilience of GLOF Reporting to Atmospheric Warming

Most glaciers in our study regions have shrunk rapidly in length, area, and mass since the 1980s (Paul & Bolch, 2019), and faster so since 2000 (Hugonnet et al., 2021), coinciding with accelerated warming that our models pick up (Figure 1c). Though variability in precipitation partly explains regional changes in glacier mass losses, modeling (Marzeion et al., 2014) and observations (Hugonnet et al., 2021) suggest that most contemporary global glacier melt is likely due to rising temperatures. Yet even enhanced warming and continued lake formation since the 1970s fail to explain the much less increasing number of yearly reported GLOFs compared to earlier periods. A regression model including temperature and research activity (Figure 3) shows that only glacier-dammed lakes in the Pacific Northwest burst out more frequently possibly associated with rising air temperatures. The number of glacier surveys is no credible predictor in any region, reflecting the generally lowered increase of these surveys since the early 1970s (Figure 3, Figure S2 in Supporting Information S1).

A simple linear relationship between GLOF occurrences and atmospheric warming thus remains elusive, at least in the past five decades, for which we have the highest data density anywhere in our study regions (Carrivick & Tweed, 2016). This missing link might be explained from a theory (Harrison et al., 2018) that expects GLOFs to lag by some decades behind warming, glacier recession, and meltwater production. Hence, different regional stages of deglaciation could blur a global trend in GLOF activity. Yet regardless of whether we place the onset of deglaciation as early as 1850 in the European Alps (Painter et al., 2013) or about one century later in High Mountain Asia (Rowan, 2017): both regions show unchanged rates of GLOF reporting despite ongoing warming in the





Figure 3. Effects of temperatures and research activity on GLOF reporting in the period 1973–2017. (a) Reported GLOFs versus mean annual air temperatures with the number of glacier surveys held constant. Thick lines are mean rates and shade is the 95% HDI from a hierarchical Poisson regression model; gray lines and tones are dam types that have no credible trends. (b) Posterior slopes (i.e., weights) of air temperatures in the model. (c) Posterior weights of glacier surveys. We standardized both predictors to measure their relative contribution to the outcome. Asterisks denote posterior distributions that are credibly different from zero (black vertical line).

period 1973–2017 (Figure S3 in Supporting Information S1). Arguably, air temperatures have a high inter-annual variance, compromising correlations with GLOF activity. Many studies thus resorted to changes in glacier lakes as a possibly more suitable diagnostic of a growing hazard potential from GLOFs (Shugar et al., 2020; Wang et al., 2020). Globally, the size and abundance of glacier lakes has doubled between 1990 and 2017 (Shugar et al., 2020). Most of the growth in glacier lake area has been tied to moraine-dammed lakes (Buckel et al., 2018; Emmer et al., 2020; Rick et al., 2022; Wilson et al., 2018; Zheng, Allen, et al., 2021), motivating numerous GLOF hazard and risk appraisals (Emmer, 2018). However, our assessment is consistent with other regional assessments that noted a lack of any long-term, continuous increase in the frequency of moraine-dam failures (Emmer et al., 2020; Veh et al., 2019). The temperature-resilience of moraine-dam failures in recent decades may indicate some stability against the growing volumes of these lakes (Lala et al., 2018). Vegetation cover may stabilize moraine slopes or large boulders can armor outflow channels (Clague & Evans, 2000). Dwindling snow packs and glaciers further reduce the potential for snow and ice avalanches to enter glacier lakes (Matiu et al., 2021; Musselman et al., 2021). Though many glacier lakes have grown toward steep rock walls prone to rockfalls and landslides (Haeberli et al., 2017; Zheng, Bao, et al., 2021), little is known about the changing frequency of such triggers. Artificial drainages of moraine-dammed lakes may also have helped to maintain a constant GLOF rate in past decades (Emmer et al., 2018; Haeberli et al., 2001). Ice-dammed lakes, in contrast, are more short-lived, and their formation is tied to thick glaciers that can impound lakes at their margins, a common situation along glaciers descending from ice fields in Alaska (Field et al., 2021; Rick et al., 2022) and Patagonia (Wilson et al., 2018). With climate-enhanced melting, ice-dammed lakes can enter cycles of outbursts such as lakes dammed by Tulsequah or Kennicott Glaciers that produced multiple GLOFs per year during the 20th century (Geertsema & Clague, 2005; Rickman & Rosenkrans, 1997). Yet several hundreds of ice-dammed lakes have disappeared in our study regions as a consequence of downwasting and retreat of glacier dams in past decades (Geertsema &



Clague, 2005; Tweed & Russell, 1999; Wolfe et al., 2014). In Alaska, for example, the total number and area of ice-dammed lakes decreased by 13% and 43%, respectively, between 1984 and 2019. Only 16 new ice-dammed lakes have reformed in this period, possibly at higher elevations in recently deglaciated tributary valleys (Rick et al., 2022). Hence the regional increases in ice-dammed failures (Figure 2) are tied to fewer lakes that burst more frequently. Despite some current positive trends, we may thus see declining trends in ice-dam failures sometime in the future, when ongoing ice loss impedes the formation of such dams (Rick et al., 2022).

3.4. Biases in GLOF Reporting in the Early 20th Century

We infer that the number of potential source locations for GLOFs has continuously changed in our study period as moraine dams have ceased to exist and glaciers have retreated. Given limited remaining evidence in the field and historic changes in research practice, we emphasize that this reporting rate differs from the actual, and unknown, rate of GLOFs. Yet, our change-point models allow us to at least approximate the total number of under-sampled GLOFs in our inventory. To this end, we hind- or forecast annual GLOF counts for either segment before and after 1973, using the linear model for GLOF count and year (Figure 2), or using the predictors "temperature" and "glaciological surveys" (Figure 4). The hindcasting models shows that the total number of GLOFs before 1973 could be on average 313% ($\hat{n} = 3,560^{+4,731}/_{-1,521}$ median and HDI using a simple hindcast without covariates) or 94% ($\hat{n} = 1,671^{+2,129}/_{-453}$ using temperature and glaciological surveys as predictors) higher, compared to the reported 860 GLOFs in that period (Figure 4a). Simply extrapolating back the recent reporting rate yields unduly high GLOF counts in Iceland and High Mountain Asia. The declining trend in reported ice- and moraine-dammed failures in recent decades (Figure 2) would thus predict much higher GLOF rates in the early 20th century (Figure 4a). The other hindcasting model predicts the highest count of potentially unobserved GLOFs in the Pacific Northwest because of large differences in reported GLOFs per unit temperature either side of the break point (Figure 4a). This model fits best in the European Alps where the 137 reported GLOFs overlap with some 169⁺⁷¹/₋₅₅ predicted GLOFs. Regardless of model choice, all hindcasts imply that GLOFs have likely been underreported for most of the 20th century outside the European Alps. Glacier-related disasters in the Alps caused hundreds of fatalities in the 19th century (Ancey et al., 2019; Haeberli, 1983; Vincent, Garambois, et al., 2010), possibly motivating the highest glaciological research activity in our study regions (Figure S2 in Supporting Information S1) and also widespread mitigation efforts to drain lakes artificially (Portocarerro, 2014; Vincent, Auclair,



Figure 4. Predicted GLOF counts per region for periods before and after the reporting break in 1973. (a) Hindcasted GLOF counts for the period 1901–1972, drawing on data from the period 1973–2017. (b) Forecast GLOF counts for the period 1973–2017, drawing on data from the period 1901–1972. We compare models that extrapolate GLOF counts either from the predictor "Year" or from the two predictors "Annual air temperature" and "Number of glacier surveys." Violins are the density of predicted GLOF counts; black horizontal lines are the 25th, 50th, and 75th percentiles. Thick blue line is the reported number of GLOFs in the associated period.

et al., 2010). Many geographical societies and geography chairs at universities were established in Europe in the 19th century, possibly contributing to the early research interest in mountains and glaciers. Elsewhere, systematic glaciological explorations began decades later (Figure S2 in Supporting Information S1). In Alaska, for example, the first glacier surveys became available in the late 1940s, while stream gages (another method to capture GLOF discharges) were installed as late as in the 1960s. This poor instrumentation and research activity is consistent with a GLOF reporting bias in this region in the early 20th century and possibly earlier (Rabot, 1905). Evidence for underreporting is also growing in High Mountain Asia and in the Northern Andes, where historic aerial photographs show characteristic deposits from GLOFs (Emmer et al., 2021; Zheng, Bao, et al., 2021). These regions likely had at least some 180 and 100 undated GLOFs, respectively, in the (cooler) early 20th century. Moreover, studies of Patagonian floodplain sediments point to elevated GLOF frequencies in the 17th–19th centuries when glaciers advanced and impounded more water in a colder and wetter climate (Benito et al., 2021; Vandekerkhove et al., 2020).

The explanatory power of warming is also limited when forecasting GLOF counts for the period 1973–2017 from data of the preceding seven decades. Using temperature and glaciological surveys as predictors, we obtain a total count of $1,162^{+298}/_{-210}$ GLOFs, matching well with the observed count of 1,105 GLOFs (Figure 4b). This model has credible weights for the predictor glaciological surveys for most dam types and regions, whereas temperature has ambiguous weights (Figure S4 in Supporting Information S1). Extrapolating solely the growing trends in GLOF reporting during the period 1901–1972 (Figure S5 in Supporting Information S1) predicts a total of $2,677^{+2.005}/_{-884}$ GLOFs, thus overestimating the reported GLOF count on average by 142% (Figure 4b). This mismatch shows that historical GLOF rates from the early 20th century alone may be unsuitable for extrapolation. The resulting over-predictions support the use of a change-point model instead that captures better the change toward a more consistent research activity in the late-20th century.

We find that potential biases in the reported frequency of GLOFs have been rarely accounted for, let alone quantified systematically in previous inventories (Carrivick & Tweed, 2016; Harrison et al., 2018; Nie et al., 2018). Our assessment resorts to few predictors to reduce effects of collinearity, and provides a first approximation of the actual rate of GLOFs, which is a fundamental requirement for estimating GLOF hazard. This model framework is flexible and extensible in a number of ways.

First, future studies may assess the role of predictors other than those we chose to distinguish better between a physical- and research-driven recording of GLOFs. For example, changes in the abundance and area of glacier lakes may be tied closer to the annual number of reported GLOFs than annual air temperatures. The growing archive of satellite images makes it possible to map the annual distribution of glacial lakes in a given mountain region, yet with the study period remaining limited to the past decade (F. Chen et al., 2021). Similar temporal constraints apply to useful predictors such as glacier area or thickness, which could track changes in lake exposure and meltwater contribution to lakes (Hugonnet et al., 2021; Shean et al., 2020). Our proxy for research activity implies that GLOF reporting hinges on field-based measurements of glaciers, in particular for the early 20th century when methods for remote detection were low. Some of the saturation in field-based glacier studies after the 1970s could be explained by the increasing role of remote sensing in the following decades (Wulder et al., 2012). Indeed, dozens of previously unrecorded GLOFs were detected in satellite imagery, especially from the Landsat program, during this period (Emmer et al., 2021; McKillop & Clague, 2007; Veh et al., 2019; Wilson et al., 2018). However, the trend in reported GLOF counts has been outpaced by the continuous growth and access to remote sensing products, and declined compared to the previous period. Useful alternative predictors for exploring possible biases in regional GLOF rates might be data on changes in population density; development in infrastructure or traffic routes; or tourist visits to high mountains. Such information is often confidential and difficult to merge owing to different data sources, compared to the wealth of more than 12 decades of standardised glacier monitoring (WGMS, 2020).

Second, our hierarchical model structure allows for testing or including other levels than those we chose. For example, GLOFs originating from bedrock-dammed or supraglacial lakes are assumed to become more frequent with ongoing glacier recession (Emmer et al., 2022; Miles et al., 2018). Those lakes can enter as additional hierarchical levels as more failures from these dam types enter our inventory. Rather than focusing on dam types, our model could also distinguish between triggers or other prominent factors that could change the stability of the dam or the adjacent basin. For example, we could group the data by metrics of changing slope stability due to glacial de-buttressing and permafrost warming or changing rates of ice avalanches or glacier calving



(Ballesteros-Cánovas et al., 2018; Deline et al., 2021; Röhl, 2006). Adding such predictors to our hierarchical models could thus help to identify regions or situations that are conducive to generating GLOFs. Finally, future appraisals may also explore potential delays between initial atmospheric forcing and the eventual occurrence of GLOFs (Harrison et al., 2018). Our current model searches for the optimal linear feedback between predictor and outcome, while lagged or autoregressive models may help to investigate notions of a lagged response to post-Little Ice Age warming (Harrison et al., 2018).

4. Conclusion

We compiled the largest and globally most consistent GLOF inventory that reveals a steep rise in reported GLOFs from the 1900s to 1970s. This rise has flattened since, despite a nearly synchronous increase in atmospheric warming rates. This divergence in trends undermines notions of a largely climate-driven occurrence of GLOFs. Alternatively, we propose that a growing research interest in glaciers largely explains global trends of reported GLOFs. The more consistent research practice since the 1970s coincides with a period of dense GLOF reporting and may be more appropriate to estimate the actual, but unknown, GLOF rate. Future studies may thus consider this period as a more robust baseline to avoid gross misestimates in trends of GLOF frequency, hazard, or risk.

Our analysis challenges the prevailing view that predicts more GLOFs to come in a warming atmosphere. Unhalted warming since the 1970s is thought to have promoted rapid glacier retreat and lake expansion, though we found little evidence for a commensurate increase in GLOF activity. While moraine-dammed lakes attract most of the current research attention, we maintain that ice-dammed lakes are the drivers of GLOF trends and abundance, given the negligible changes in reported GLOFs for all other dam types. Most historic ice-dam failures have happened in regions with extensive glacier cover such as Alaska, Patagonia, and the Karakoram. Anticipating GLOF occurrences and impacts in the immediate future may thus focus more on predicting locations suitable for the formation of ice dams. We emphasize that it is vital to distinguish better between physical and research-oriented drivers of regional GLOF trends or realistic assessments of changing hazard potential and related and risk under future atmospheric warming. Such appraisals will help to prevent and mitigate some of the damages and losses that GLOFs have repeatedly brought to communities in high mountains.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

All codes for statistical analysis and figure production are available at https://doi.org/10.5281/zenodo.5503969. The GLOF inventory is available at http://glofs.geoecology.uni-potsdam.de.

References

- Ancey, C., Bardou, E., Funk, M., Huss, M., Werder, M. A., & Trewhela, T. (2019). Hydraulic reconstruction of the 1818 Giétro glacial lake outburst flood. Water Resources Research, 55(11), 8840–8863. https://doi.org/10.1029/2019WR025274
- Bajracharya, S. R., & Mool, P. (2009). Glaciers, glacial lakes and glacial lake outburst floods in the Mount Everest region, Nepal. Annals of Glaciology, 50(53), 81–86. https://doi.org/10.3189/172756410790595895
- Ballesteros-Cánovas, J. A., Trappmann, D., Madrigal-González, J., Eckert, N., & Stoffel, M. (2018). Climate warming enhances snow avalanche risk in the Western Himalayas. *Proceedings of the National Academy of Sciences*, 115(13), 3410–3415. https://doi.org/10.1073/ pnas.1716913115
- Baťka, J., Vilímek, V., Štefanová, E., Cook, S. J., & Emmer, A. (2020). Glacial Lake outburst floods (GLOFs) in the Cordillera Huayhuash, Peru: Historic events and current susceptibility. *Water*, 12(10), 2664. https://doi.org/10.3390/w12102664
- Benito, G., Thorndycraft, V. R., Medialdea, A., Machado, M. J., Sancho, C., & Dussaillant, A. (2021). Declining discharge of glacier outburst floods through the Holocene in central Patagonia. *Quaternary Science Reviews*, 256, 106810. https://doi.org/10.1016/j.quascirev.2021.106810

Bhambri, R., Hewitt, K., Kawishwar, P., Kumar, A., Verma, A., Snehmani, et al. (2019). Ice-dams, outburst floods, and movement heterogeneity of glaciers, Karakoram. *Global and Planetary Change*, 180, 100–116. https://doi.org/10.1016/j.gloplacha.2019.05.004

- Bolch, T., Kulkarni, A., Kääb, A., Huggel, C., Paul, F., Cogley, J. G., et al. (2012). The state and fate of Himalayan glaciers. *Science*, 336(6079), 310–314. https://doi.org/10.1126/science.1215828
- Buckel, J., Otto, J. C., Prasicek, G., & Keuschnig, M. (2018). Glacial lakes in Austria Distribution and formation since the Little Ice Age. Global and Planetary Change, 164, 39–51. https://doi.org/10.1016/j.gloplacha.2018.03.003
- Bürkner, P.-C. (2017). An R package for bayesian multilevel models using Stan. Journal of Statistical Software, 80(1). https://doi.org/10.18637/jss.v080.i01

Acknowledgments

We thank Victor Blagoveshchensky, Jonathan Carrivick, Bergur Einarsson, Pablo Iribarren Anacona, Marten Geertsema, Matthias Huss, Eric Parra Hormazábal, Miriam Jackson, Jakob Steiner, Fiona Tweed, Vanessa Round, Mauro Werder, and Vanessa Winchester for sharing data on historic lake outbursts. Two reviewers helped us improve the manuscript. Open Access funding enabled and organized by Projekt DEAL.

- Carrivick, J. L., & Tweed, F. S. (2016). A global assessment of the societal impacts of glacier outburst floods. *Global and Planetary Change*, 144, 1–16. https://doi.org/10.1016/j.gloplacha.2016.07.001
- Chen, F., Zhang, M., Guo, H., Allen, S., Kargel, J. S., Haritashya, U. K., & Watson, C. S. (2021). Annual 30 m dataset for glacial lakes in High Mountain Asia from 2008 to 2017. Earth System Science Data, 13(2), 741–766. https://doi.org/10.5194/essd-13-741-2021
- Chen, Y. (2010). Response of glacial-lake outburst floods to climate change in the Yarkant River basin on northern slope of Karakoram Mountains, China. *Quaternary International*, 7. https://doi.org/10.1016/j.quaint.2010.01.003
- Clague, J., & Evans, S. G. (2000). A review of catastrophic drainage of moraine-dammed lakes in British Columbia. *Quaternary Science Reviews*, 19(17–18), 1763–1783. https://doi.org/10.1016/S0277-3791(00)00090-1
- Clarke, G. K. C. (2003). Hydraulics of subglacial outburst floods: New insights from the spring–Hutter formulation. Journal of Glaciology, 49(165), 299–313. https://doi.org/10.3189/172756503781830728
- Cook, K. L., Andermann, C., Gimbert, F., Adhikari, B. R., & Hovius, N. (2018). Glacial lake outburst floods as drivers of fluvial erosion in the Himalaya. Science, 362(6410), 53–57. https://doi.org/10.1126/science.aat4981
- Deline, P., Gruber, S., Amann, F., Bodin, X., Delaloye, R., Failletaz, J., et al. (2021). Ice loss from glaciers and permafrost and related slope instability in high-mountain regions. In W. Haeberli & C. Whiteman (Eds.), Snow and ice-related hazards, risks, and disasters (pp. 501–540). Elsevier. https://doi.org/10.1016/b978-0-12-817129-5.00015-9
- Emmer, A. (2017). Geomorphologically effective floods from moraine-dammed lakes in the Cordillera Blanca, Peru. Quaternary Science Reviews, 177, 220–234. https://doi.org/10.1016/j.quascirev.2017.10.028
- Emmer, A. (2018). GLOFs in the WOS: Bibliometrics, geographies and global trends of research on glacial lake outburst floods (Web of Science, 1979–2016). Natural Hazards and Earth System Sciences, 18(3), 813–827. https://doi.org/10.5194/nhess-18-813-2018
- Emmer, A., Cook, S., Wood, J. L., Harrison, S., Wilson, R., Diaz-Moreno, A., et al. (2021). A new GLOF inventory for the Peruvian and Bolivian Andes. Presented at the EGU General Assembly, 2021. https://doi.org/10.5194/egusphere-egu21-9744
- Emmer, A., Harrison, S., Mergili, M., Allen, S., Frey, H., & Huggel, C. (2020). 70 years of lake evolution and glacial lake outburst floods in the Cordillera Blanca (Peru) and implications for the future. *Geomorphology*, 365, 107178. https://doi.org/10.1016/j.geomorph.2020.107178 Emmer, A., & Vilímek, V. (2013). Review article: Lake and breach hazard assessment for moraine-dammed lakes: An example from the Cordil-
- lera Blanca (Peru). Natural Hazards and Earth System Sciences, 13(6), 1551–1565. https://doi.org/10.5194/nhess-13-1551-2013 Emmer, A., Vilímek, V., & Zapata, M. L. (2018). Hazard mitigation of glacial lake outburst floods in the Cordillera Blanca (Peru): The effective-
- ness of remedial works: Hazard mitigation of GLOFs in the Cordillrea Blanca. Journal of Flood Risk Management, 11, S489–S501. https:// doi.org/10.1111/jfr3.12241
- Emmer, A., Wood, J. L., Cook, S. J., Harrison, S., Wilson, R., Diaz-Moreno, A., et al. (2022). 160 glacial lake outburst floods (GLOFs) across the Tropical Andes since the Little Ice Age. *Global and Planetary Change*, 208, 103722. https://doi.org/10.1016/j.gloplacha.2021.103722
- Field, H. R., Armstrong, W. H., & Huss, M. (2021). Gulf of Alaska ice-marginal lake area change over the Landsat record and potential physical controls. *The Cryosphere*, 15(7), 3255–3278. https://doi.org/10.5194/tc-15-3255-2021
- Geertsema, M., & Clague, J. J. (2005). Jökulhlaups at Tulsequah Glacier, northwestern British Columbia, Canada. *The Holocene*, 15(2), 310–316. https://doi.org/10.1191/0959683605hl812rr
- Gudmundsson, M. T., Björnsson, H., & Pálsson, F. (1995). Changes in Jökulhlaup sizes in Grímsvötn, Vatnajökull, Iceland, 1934-91, deduced from in-situ measurements of subglacial lake volume. *Journal of Glaciology*, 41(138), 263–272. https://doi.org/10.3189/S0022143000016166
- Guðmundsson, S., Björnsson, H., Pálsson, F., Magnússon, E., Sæmundsson, P., & Jóhannesson, T. (2020). Terminus lakes on the south side of Vatnajökull ice cap, SE-Iceland. Jökull Journal, 69, 1–34. https://doi.org/10.33799/jokull2019.69.001
- Haeberli, W. (1983). Frequency and characteristics of glacier floods in the Swiss Alps. Annals of Glaciology, 4, 85–90. https://doi.org/10.3189/ S0260305500005280
- Haeberli, W., Kääb, A., Mühll, D. V., & Teysseire, P. (2001). Prevention of outburst floods from periglacial lakes at Grubengletscher, Valais, Swiss Alps. Journal of Glaciology, 47(156), 111–122. https://doi.org/10.3189/172756501781832575
- Haeberli, W., Schaub, Y., & Huggel, C. (2017). Increasing risks related to landslides from degrading permafrost into new lakes in de-glaciating mountain ranges. *Geomorphology*, 293, 405–417. https://doi.org/10.1016/j.geomorph.2016.02.009
- Harris, I., Osborn, T. J., Jones, P., & Lister, D. (2020). Version 4 of the CRU TS monthly high-resolution gridded multivariate climate dataset. Scientific Data, 7(1), 109. https://doi.org/10.1038/s41597-020-0453-3
- Harrison, S., Kargel, J. S., Huggel, C., Reynolds, J., Shugar, D. H., Betts, R. A., et al. (2018). Climate change and the global pattern of morainedammed glacial lake outburst floods. *The Cryosphere*, 12(4), 1195–1209. https://doi.org/10.5194/tc-12-1195-2018
- Hugonnet, R., McNabb, R., Berthier, E., Menounos, B., Nuth, C., Girod, L., et al. (2021). Accelerated global glacier mass loss in the early twenty-first century. *Nature*, 592(7856), 726–731. https://doi.org/10.1038/s41586-021-03436-z
- Huss, M., Bookhagen, B., Huggel, C., Jacobsen, D., Bradley, R. S., Clague, J. J., et al. (2017). Toward mountains without permanent snow and ice. *Earth's Future*, 5(5), 418–435. https://doi.org/10.1002/2016EF000514
- Immerzeel, W. W., Lutz, A. F., Andrade, M., Bahl, A., Biemans, H., Bolch, T., et al. (2020). Importance and vulnerability of the world's water towers. *Nature*, 577(7790), 364–369. https://doi.org/10.1038/s41586-019-1822-y
- Iribarren Anacona, P., Mackintosh, A., & Norton, K. P. (2015). Hazardous processes and events from glacier and permafrost areas: Lessons from the Chilean and Argentinean Andes. *Earth Surface Processes and Landforms*, 40(1), 2–21. https://doi.org/10.1002/esp.3524
- Iturrizaga, L. (2011). Glacier lake outburst floods. In V. P. Singh, P. Singh, & U. K. Haritashya (Eds.), Encyclopedia of snow, ice and glaciers (pp. 381–399). Springer Netherlands. https://doi.org/10.1007/978-90-481-2642-2_196
- Jacquet, J., McCoy, S. W., McGrath, D., Nimick, D. A., Fahey, M., O'kuinghttons, J., et al. (2017). Hydrologic and geomorphic changes resulting from episodic glacial lake outburst floods: Rio Colonia, Patagonia, Chile. *Geophysical Research Letters*, 44(2), 854–864. https://doi. org/10.1002/2016GL071374
- King, O., Bhattacharya, A., Bhambri, R., & Bolch, T. (2019). Glacial lakes exacerbate Himalayan glacier mass loss. Scientific Reports, 9(1), 18145. https://doi.org/10.1038/s41598-019-53733-x
- Kingslake, J., & Ng, F. (2013). Quantifying the predictability of the timing of Jökulhlaups from Merzbacher Lake, Kyrgyzstan. Journal of Glaciology, 59(217), 805–818. https://doi.org/10.3189/2013JoG12J156
- Lala, J. M., Rounce, D. R., & McKinney, D. C. (2018). Modeling the glacial lake outburst flood process chain in the Nepal Himalaya: Reassessing Imja Tsho's hazard. *Hydrology and Earth System Sciences*, 22(7), 3721–3737. https://doi.org/10.5194/hess-22-3721-2018
- Lindeløv, J. K. (2020). mcp: An R package for regression with multiple change points (preprint). Open Science Framework. https://doi. org/10.31219/osf.io/fzqxv
- Liu, J.-J., Cheng, Z.-L., & Su, P.-C. (2014). The relationship between air temperature fluctuation and Glacial Lake Outburst Floods in Tibet, China. Quaternary International, 321, 78–87. https://doi.org/10.1016/j.quaint.2013.11.023

- Magnússon, E., Gudmundsson, M. T., Roberts, M. J., Sigurðsson, G., Höskuldsson, F., & Oddsson, B. (2012). Ice-volcano interactions during the 2010 Eyjafjallajökull eruption, as revealed by airborne imaging radar. *Journal of Geophysical Research*, 117(B7). https://doi. org/10.1029/2012JB009250
- Marzeion, B., Cogley, J. G., Richter, K., & Parkes, D. (2014). Attribution of global glacier mass loss to anthropogenic and natural causes. *Science*, 345(6199), 919–921. https://doi.org/10.1126/science.1254702
- Matiu, M., Crespi, A., Bertoldi, G., Carmagnola, C. M., Marty, C., Morin, S., et al. (2021). Observed snow depth trends in the European Alps: 1971 to 2019. *The Cryosphere*, 15(3), 1343–1382. https://doi.org/10.5194/tc-15-1343-2021
- McKillop, R. J., & Clague, J. J. (2007). Statistical, remote sensing-based approach for estimating the probability of catastrophic drainage from moraine-dammed lakes in southwestern British Columbia. *Global and Planetary Change*, 56(1–2), 153–171. https://doi.org/10.1016/j. gloplacha.2006.07.004
- Miles, E. S., Watson, C. S., Brun, F., Berthier, E., Esteves, M., Quincey, D. J., et al. (2018). Glacial and geomorphic effects of a supraglacial lake drainage and outburst event, Everest region, Nepal Himalaya. *The Cryosphere*, *12*(12), 3891–3905. https://doi.org/10.5194/tc-12-3891-2018
 Milner, A. M., Khamis, K., Battin, T. J., Brittain, J. E., Barrand, N. E., Füreder, L., et al. (2017). Glacier shrinkage driving global changes in
- Minier, A. M., Knahns, K., Battin, I. J., Britani, J. E., Barrand, N. E., Furduer, L., et al. (2017). Gracier similarge unving global changes in downstream systems. *Proceedings of the National Academy of Sciences*, 114(37), 9770–9778. https://doi.org/10.1073/pnas.1619807114
- Mountain Research Initiative EDW Working Group. (2015). Elevation-dependent warming in mountain regions of the world. *Nature Climate Change*, 5(5), 424–430. https://doi.org/10.1038/nclimate2563
- Musselman, K. N., Addor, N., Vano, J. A., & Molotch, N. P. (2021). Winter melt trends portend widespread declines in snow water resources. *Nature Climate Change*, 11(5), 418–424. https://doi.org/10.1038/s41558-021-01014-9
- Ng, F., & Liu, S. (2009). Temporal dynamics of a jökulhlaup system. *Journal of Glaciology*, 55(192), 651–665. https://doi.org/10.3189/002214309789470897
- Nie, Y., Liu, Q., Wang, J., Zhang, Y., Sheng, Y., & Liu, S. (2018). An inventory of historical glacial lake outburst floods in the Himalayas based on remote sensing observations and geomorphological analysis. *Geomorphology*, 308, 91–106. https://doi.org/10.1016/j.geomorph.2018.02.002
- Painter, T. H., Flanner, M. G., Kaser, G., Marzeion, B., VanCuren, R. A., & Abdalati, W. (2013). End of the Little Ice Age in the Alps forced by industrial black carbon. *Proceedings of the National Academy of Sciences*, 110(38), 15216–15221. https://doi.org/10.1073/pnas.1302570110 Paul, F., & Bolch, T. (2019). Glacier changes since the little ice age. In T. Heckmann & D. Morche (Eds.), *Geomorphology of proglacial systems*
- (pp. 23–42). Springer. https://doi.org/10.1007/978-3-319-94184-4_2 Pfeffer, W. T., Arendt, A. A., Bliss, A., Bolch, T., Cogley, J. G., Gardner, A. S., et al. (2014). The Randolph Glacier inventory: A globally complete
- inventory of glaciers. Journal of Glaciology, 60(221), 537–552. https://doi.org/10.3189/2014JoG13J176 Portocarerro, C. (2014). The glacial lake handbook: Reducing risk from dangerous Glacial Lakes in the cordillera Blanca, Peru (tech-
- nical report). United States Agency for International Development. Retrieved from https://www.climatelinks.org/resources/ glacial-lake-handbook-reducing-risk-dangerous-glacial-lakes-cordillera-blanca-peru

Rabot, C. (1905). Glacial reservoirs and their outbursts. The Geographical Journal, 25(5), 534. https://doi.org/10.2307/1776694

- Richardson, S. D., & Reynolds, J. M. (2000). An overview of glacial hazards in the Himalayas. *Quaternary International*, 65–66, 31–47. https://doi.org/10.1016/S1040-6182(99)00035-X
- Rick, B., McGrath, D., Armstrong, W., & McCoy, S. W. (2022). Dam type and lake location characterize ice-marginal lake area change in Alaska and NW Canada between 1984 and 2019. The Cryosphere, 16, 297–314. https://doi.org/10.5194/tc-16-297-2022
- Rickman, R. L., & Rosenkrans, D. S. (1997). Hydrologic conditions and hazards in the Kennicott River Basin, Wrangell-St. Elias National Park Preserve, Alaska. Report No. 96–4296. https://doi.org/10.3133/wri964296
- Röhl, K. (2006). Thermo-erosional notch development at fresh-water-calving Tasman Glacier, New Zealand. Journal of Glaciology, 52(177), 203–213. https://doi.org/10.3189/172756506781828773
- Rowan, A. V. (2017). The 'Little Ice Age' in the Himalaya: A review of glacier advance driven by Northern Hemisphere temperature change. *The Holocene*, 27(2), 292–308. https://doi.org/10.1177/0959683616658530
- Schwanghart, W., Worni, R., Huggel, C., Stoffel, M., & Korup, O. (2016). Uncertainty in the Himalayan energy–water nexus: Estimating regional exposure to glacial lake outburst floods. *Environmental Research Letters*, 11(7), 074005. https://doi.org/10.1088/1748-9326/11/7/074005
- Shean, D. E., Bhushan, S., Montesano, P., Rounce, D. R., Arendt, A., & Osmanoglu, B. (2020). A systematic, regional assessment of high mountain Asia glacier mass balance. *Frontiers of Earth Science*, 7, 363. https://doi.org/10.3389/feart.2019.00363
- Shugar, D. H., Burr, A., Haritashya, U. K., Kargel, J. S., Watson, C. S., Kennedy, M. C., et al. (2020). Rapid worldwide growth of glacial lakes since 1990. Nature Climate Change, 10(10), 939–945. https://doi.org/10.1038/s41558-020-0855-4
- Tweed, F. S., & Russell, A. J. (1999). Controls on the formation and sudden drainage of glacier-impounded lakes: Implications for jökulhlaup characteristics. Progress in Physical Geography, 23(1), 79–110. https://doi.org/10.1177/030913339902300104
- Vandekerkhove, E., Bertrand, S., Mauquoy, D., McWethy, D., Reid, B., Stammen, S., et al. (2020). Neoglacial increase in high-magnitude glacial lake outburst flood frequency, upper Baker River, Chilean Patagonia (47°S). *Quaternary Science Reviews*, 248, 106572. https://doi. org/10.1016/j.quascirev.2020.106572
- Veh, G., Korup, O., Roessner, S., & Walz, A. (2018). Detecting Himalayan glacial lake outburst floods from Landsat time series. *Remote Sensing of Environment*, 207, 84–97. https://doi.org/10.1016/j.rse.2017.12.025
- Veh, G., Korup, O., Spechtvon, S., Roessner, S., & Walz, A. (2019). Unchanged frequency of moraine-dammed glacial lake outburst floods in the Himalaya. *Nature Climate Change*, 9(5), 379–383. https://doi.org/10.1038/s41558-019-0437-5
- Vehtari, A., Gelman, A., & Gabry, J. (2017). Practical Bayesian model evaluation using leave-one-out cross-validation and WAIC. Statistics and Computing, 27(5), 1413–1432. https://doi.org/10.1007/s11222-016-9696-4
- Vincent, C., Auclair, S., & Meur, E. L. (2010). Outburst flood hazard for glacier-dammed Lac de Rochemelon, France. Journal of Glaciology, 56(195), 91–100. https://doi.org/10.3189/002214310791190857
- Vincent, C., Garambois, S., Thibert, E., Lefebvre, E., Meur, E. L., & Six, D. (2010). Origin of the outburst flood from Glacier de Tête rousse in 1892 (Mont Blanc area, France). Journal of Glaciology, 56(198), 688–698. https://doi.org/10.3189/002214310793146188
- Wang, X., Guo, X., Yang, C., Liu, Q., Wei, J., Zhang, Y., et al. (2020). Glacial lake inventory of High Mountain Asia (1990-2018) derived from Landsat images. *Earth System Science Data*, 12, 2169–2182. https://doi.org/10.5194/essd-12-2169-2020
- WGMS. (2020). Global glacier change bulletin No. 3 (2016–2017) [Dataset]. World Glacier Monitoring Service (WGMS). https://doi.org/10.5904/ WGMS-FOG-2019-12
- Wikstrom Jones, K., & Wolken, G. J. (2019). Valdez Glacier ice-dammed lake: June 2018 glacial lake outburst flood (No. PIR 2019-4). Alaska Division of Geological & Geophysical Surveys. https://doi.org/10.14509/30175
- Wilson, R., Glasser, N. F., Reynolds, J. M., Harrison, S., Anacona, P. I., Schaefer, M., & Shannon, S. (2018). Glacial lakes of the Central and Patagonian Andes. *Global and Planetary Change*, *162*, 275–291. https://doi.org/10.1016/j.gloplacha.2018.01.004

- Wolfe, D. F. G., Kargel, J. S., & Leonard, G. J. (2014). Glacier-dammed ice-marginal lakes of Alaska. In J. S. Kargel, G. J. Leonard, M. P. Bishop, A. Kääb, & B. H. Raup (Eds.), Global land ice measurements from space (pp. 263–295). Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-540-79818-7_12
- Wulder, M. A., Masek, J. G., Cohen, W. B., Loveland, T. R., & Woodcock, C. E. (2012). Opening the archive: How free data has enabled the science and monitoring promise of Landsat. *Remote Sensing of Environment*, 122, 2–10. https://doi.org/10.1016/j.rse.2012.01.010
- Zemp, M., Huss, M., Thibert, E., Eckert, N., McNabb, R., Huber, J., et al. (2019). Global glacier mass changes and their contributions to sea-level rise from 1961 to 2016. *Nature*, 568(7752), 382–386. https://doi.org/10.1038/s41586-019-1071-0
- Zheng, G., Allen, S. K., Bao, A., Ballesteros-Cánovas, J. A., Huss, M., Zhang, G., et al. (2021). Increasing risk of glacial lake outburst floods from future Third Pole deglaciation. *Nature Change*, 11(5), 411–417. https://doi.org/10.1038/s41558-021-01028-3
- Zheng, G., Bao, A., Allen, S., Antonio Ballesteros-Cánovas, J., Yuan, Y., Jiapaer, G., & Stoffel, M. (2021). Numerous unreported glacial lake outburst floods in the Third Pole revealed by high-resolution satellite data and geomorphological evidence. *Science Bulletin*, S2095927321000505. https://doi.org/10.1016/j.scib.2021.01.014